Author's response to reviews

Title: Patient Advocacy: Barriers and Facilitators

Authors:

Reza Negarandeh (negarandeh@gmail.com)
Fatemeh Oskouie (nmsi@iums.ac.ir)
Fazlollah Ahmadi (fazlollahah@yahoo.com)
Mansoure Nikravesh (manynik@yahoo.com)
Ingilill Rahm Hallberg (Ingilill.Rahm_Hallberg@med.lu.se)

Version: 6 Date: 14 February 2006

Author's response to reviews:

We made formatting changes in manuscript according to your request.